Author’s response to reviews

Title: Clinical Experience with Tigecycline in the Treatment of Hospital-Acquired Pneumonia Caused by Multidrug Resistant Acinetobacter baumannii

Authors:

Yangang Zhou (zhouyangang@csu.edu.cn)
Xumin Chen (1120340591@qq.com)
Ping Xu (xuping1109@csu.edu.cn)
Yan Zhu (zhuyan0918@csu.edu.cn)
Kuangguo Wang (aledith@126.com)
Daxiong Xiang (xiangdaxiong@163.com)
Feng Wang (roitech@126.com)
Hoan Banh (Hoan@ualberta.ca)

Version: 7 Date: 05 Apr 2019

Author’s response to reviews:

Dear editors,

Thank you very much for your feedback on “Clinical Experience with Tigecycline in the Treatment of Hospital-Acquired Pneumonia Caused by Multidrug Resistant Acinetobacter baumannii”, I am very appreciated about the suggestion you made. The manuscript was revised in accordance with BioMed Central editorial policies and formatting guidelines. Here are the responses to the reviewers.

1. We note that many paragraphs in the Methods section have been reused from one of your previous publication: https://doi.org/10.1016/j.jinf.2011.08.001

While we understand that the methodology is the same, we would appreciate if you try to minimise direct repetition of large sections of previously published text.

Response: We have revised most of the structures and sentences of the “methods parts” as you suggested to minimizing the overlap.